# "Policy options for the promotion of electric vehicles: a review"

Jordi Perdiguero and Juan Luis Jiménez



Institut de Recerca en Economia Aplicada | Research Institute of Applied Economics

## 

Institut de Recerca en Economia Aplicada Regional i Pública Research Institute of Applied Economics

### Universitat de Barcelona

Av. Diagonal, 690 • 08034 Barcelona

WEBSITE: www.ub.edu/irea/ • CONTACT: irea@ub.edu

The Research Institute of Applied Economics (IREA) in Barcelona was founded in 2005, as a research institute in applied <u>economics</u>. Three consolidated research groups make up the institute: AQR, RISK and GiM, and a large number of members are involved in the Institute. IREA focuses on four priority lines of investigation: (i) the quantitative study of regional and urban economic activity and analysis of regional and local economic policies, (ii) study of public economic activity in markets, particularly in the fields of empirical evaluation of privatization, the regulation and competition in the markets of public services using state of industrial economy, (iii) risk analysis in finance and insurance, and (iv) the development of micro and macro econometrics applied for the analysis of economic activity, particularly for quantitative evaluation of public policies.

IREA Working Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. For that reason, IREA Working Papers may not be reproduced or distributed without the written consent of the author. A revised version may be available directly from the author.

Any opinions expressed here are those of the author(s) and not those of IREA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions.

#### Abstract

The upward trend in fuel prices and the desire to reduce pollution levels mean that the electric vehicle has become an increasingly attractive alternative in recent years. The aim of this study is to examine the main barriers that the electric vehicle must overcome if it is to become a successful mode of transport and to review the main public policies that governments might implement to help in overcoming these obstacles. Public policies have been directed at four basic features of the electric vehicle: the charging network; increasing demand for these vehicles; industrialization and research and development programs; and the introduction of electric vehicles in programs of sustainable mobility. This article describes the public policies that have been implemented around the world to overcome the barriers to the adoption of electric vehicle so that it might become the vehicle of the future.

*JEL classification:* H23; P28; Q42. *Keywords*: Electric vehicle; public policies; recharge system.

Jordi Perdiguero, Departament de Política Econòmica. Grup de Recerca en Governs i Mercats (GiM). Institut de Recerca en Economia Aplicada (IREA). Avda. Diagonal 690. 08034. Barcelona. E-mail: jordi.perdiguero@ub.edu

Juan Luis Jiménez, Universidad de Las Palmas de Gran Canaria. Departamento de Análisis Económico Aplicado. Despacho D. 2-12. Campus de Tafira. 35017. Las Palmas. E-mail: <u>iljimenez@daea.ulpgc.es</u>; tlf: +34 928458191

Acknowledgements:

Thanks are due to Miquel Cruz for his useful comments and suggestions. The usual disclaimer applies.

#### 1. Introduction

Although electric vehicles have experienced increasing popularity in recent years, their history can in fact be traced back many years. The first electric vehicle was built in 1834 and by the early 1890s electric vehicles were on sale as one of the main competitors to the internal combustion engine (hereafter ICE), capturing more than a third of the market (Kley et al, 2011). However, ICE vehicles managed to impose themselves owing to their lower costs, both in terms of vehicle production and the price of petroleum fuel, while electric vehicles were handicapped by a travel range that was dependent on battery life.

However, rising oil prices<sup>1</sup>, the reduction in the cost and the increased autonomy of electric batteries, and increasing concern regarding the emission of greenhouse gases in developed economies<sup>2</sup>, have combined to raise interest in the potential of electric vehicles.

Thus, once again the electric vehicle has emerged as a serious competitor to the ICE vehicle. The electric vehicle offers many advantages, which can be summed up in its overall greater energy efficiency. This perhaps is the primary motive prompting public authorities to implement a package of measures to help introduce the electric vehicle at a range of levels (state, regional and local).

These measures resort not only to the traditional instrument of the subsidy for the purchase of electric vehicles, as applied to hybrid or "green" ICE vehicles in countries such as the United States, Germany, the United Kingdom and Spain<sup>3</sup>, but also to other measures that tackle the "chicken or the egg" dilemma: there are no electric cars because there are no recharging points and there are no charging points because there are no electric vehicles. Thus, additional measures have been designed by the

<sup>&</sup>lt;sup>1</sup> Prices of Brent crude oil (taking into account inflation) increased from 23.95 dollars in 1973 to 102.61 dollars in 1979 (an increase of 428.4 percent). In recent years the price increase has been equally remarkable rising from 28.59 dollars in 2002 to 87.33 dollars in 2008 (an increase of 305.5 percent).

<sup>&</sup>lt;sup>2</sup> Examples include successive Climate Conference Summits, and the European Union itself, in which transportation and vehicle markets as well as energy providers are regulated. Directives establish maximum contamination levels, including a 20% reduction in greenhouse gas emissions on 1990 levels to be achieved by 2020. These policies all play a role in stimulating and creating conditions for road transport electrification.

<sup>&</sup>lt;sup>3</sup> Jiménez et al (2012a) summarize the work in this area and analyze the effect of Spain's vehicle replacement plan on prices, sales and the environment.

public authorities to introduce recharging systems that can support the development of electric vehicles; to promote the necessary research and development; and, finally, to include these vehicles in sustainable mobility plans.

The aim of this paper is to review the broad range of measures that the public sector might implement to promote the electric vehicle and to report on the findings of the first pilot programs implemented in various countries. To the best of our knowledge, this is the first time that the public policy measures undertaken to promote the electric vehicle have been presented in a systematic and orderly way. It should be noted that in the discussion that follows we consider measures proposed to promote both electric vehicles and plug-in hybrid electric vehicles<sup>.4</sup>

The rest of the paper is organized as follows. Following on from this introduction, section 2 presents the strengths and weaknesses of electric vehicles in comparison with those of the internal combustion engine. Section 3 reports the main actions taken to date in deploying charging networks, while section 4 discusses measures taken in relation to promote the demand for electric vehicles. Measures to enhance the electric vehicle industry, primarily as part of research and development (R&D) programs, are presented in section 5, while section 6 examines those directed at incorporating electric vehicles in sustainable mobility plans. Section 7 reviews less frequently adopted initiatives, and finally the main conclusions are presented in the last section.

#### 2. Drivers of electric vehicles

Electric vehicles have a number of advantages over ICE vehicles, yet present a number of drawbacks. One of the main reasons for promoting electric vehicles is that they exploit more energy efficient technology than that use by ICE vehicles. As the WWF (2008) and the International Energy Agency (2008) report, electric vehicles are four times more energy efficient than ICE vehicles. In fact, Ahman (2001) shows how vehicles powered by alternative energy (basically, electric vehicles, plug-in hybrid electric vehicles and fuel-cell electric vehicles) are twice as energy efficient as current ICE vehicles.

<sup>&</sup>lt;sup>4</sup> Srivastava et al (2010) report that the plug-in hybrid vehicle is the next candidate for replacing existing ICE vehicles. The plug-in hybrid vehicle can serve as the bridge between ICE vehicles and full electric vehicles.

This increased energy efficiency can also result in a reduction in greenhouse gas emissions, although the magnitude of these reductions depends critically on the technology used to produce the electricity. If most of the electricity can be produced using sources of renewable energy (solar, wind, etc.), reductions in greenhouse gases will be high, but if the dominant technologies are coal and oil, the reduction will be minimal (Transport and Environment, 2010).<sup>5</sup> As Hadley and Tsvetkova (2009) point out, the impact of the introduction of hybrid or electric vehicles will depend on the technological mix used for electricity generation, as well as the time of day (demand peaks or valleys) when recharging takes place, among others.

A further advantage of the electric vehicle is that it should lead to an improvement in city air quality (as electricity generating plants are typically located some distance away) and noise levels. As Greenpeace et al (2010) report, if the increase in electricity demand coincides with a valley in consumption, this could improve the efficiency of electricity generating plants.<sup>6</sup>

Despite the potential benefits of electric cars, it should be borne in mind that they do not represent an unequivocal panacea. As the Transport & Environment report (2009b) stresses, it is likely that technological advances in electric vehicle development will not be fast enough to ensure compliance with the greenhouse gas limits fixed for the coming decades. Kageson (2005) expresses his doubts about the possibilities of hybrid vehicles being introduced quickly enough, since in 2004 only 8,500 new hybrid vehicles were registered, representing just 0.06 percent of new vehicles in EU-15.

One explanation for the slow introduction of electric vehicles is the obstacles this technology faces when compared to internal combustion. According to the Citi report (2009), the main obstacles are:

<sup>&</sup>lt;sup>5</sup> Bradley and Quinn (2010) also point to the need to change utility factors if we consider that hybrid vehicles will introduce different characteristics, different driving behaviors, in addition to other factors that can have a significant influence on the utility factor. For the Spanish case, see the analysis provided by Romero (2012).

<sup>&</sup>lt;sup>6</sup> As Ryan et al (2009) show, it is not only the introduction of electric vehicles that can improve the level of emissions but also an increase in taxes on gasoline, as the latter would help reduce the level of pollutant emissions from the fleet of vehicles. For a comprehensive review of tax effects on levels of contamination see Sterner (2007).

1. Costs. - Although the long-term costs of electric vehicles are not as great as those of ICE vehicles (based on lower maintenance and fuel costs), the cost of acquisition remains higher because of the price of the cell battery pack. Ensuring a competitive purchase price will, therefore, largely depend on the evolution of battery costs (the main cost involved in these vehicles). Predictions of battery costs vary from company to company, but seem to provide for a significant reduction, which should facilitate their competitiveness.

However, for the time being, and for some time to come, the cost of the battery will remain one of the main obstacles to the adoption of the electric vehicle, so much so that some companies are beginning to spread the cost of the battery, which is being granted under lease. The cost of acquisition seems to be a barrier to the spread of electric vehicles, and has led to public sector intervention through subsidies for the purchase of such vehicles, and to R&D support to reduce battery costs.

2. Infrastructure for recharging. - Although in some cities, such as London, Rome and Berlin, small networks exist for recharging vehicles, the spread of such networks is slow. Charging points installed in homes are slow but relatively inexpensive (around \$250), while more rapid charging requires an investment of several thousand euros. The failure to develop recharging networks can induce "range anxiety" in vehicle owners, that is, the fear of not reaching a charging point before the battery dies. This fear can be a significant barrier to the introduction of the electric vehicle, and here the public sector can play an active role in disseminating information about the location of these charging points to help reduce this "anxiety".

A further point to note regarding recharging points is their compatibility. The homogeneity of systems between countries, in order to avoid any incompatibility, is essential for the diffusion of electric vehicles. Here, there is an obvious role for public regulation.

3. Consumer acceptance. - Various reports conclude that consumers would be willing to make the switch if the electric vehicle reduced energy costs. Pike Research (2009) reports that two-thirds of consumers would even be willing to pay a higher price for the vehicle, under this condition. Thus, a regulatory framework and a set of clear, stable electricity rates are important in ensuring consumers are fully informed of the savings in their energy costs. Measures to facilitate public information concerning energy supply to the potential consumers of electric vehicles would therefore help in the introduction of electric vehicles.

4. The evolution of other technologies. - The existence of vehicles using other technologies (fuel cell, biofuels, ethanol, hydrogen, etc.) and the conversion of ICE vehicles in more environmentally friendly cars (with higher levels of fuel efficiency) represent obvious competitors for electric vehicles.<sup>7</sup> Identifying the best technology for the future and focusing public efforts in developing this technology will not be a straightforward matter.

In attempts to overcome these barriers, various factors will come into play. These can be classified as being either endogenous (government support, industry initiatives) or exogenous (increases in fuel prices, economic crisis, reduction of fossil fuel reserves) in nature. In this study, we focus above all on the former, but we must not forget the existence of the latter, which may have a significant influence on the adoption or otherwise of electric vehicles. In the case of the endogenous factors, it is not only public authorities that can promote the introduction of electric vehicles but industry too has a role to play in overcoming the barriers that hinder development (especially of batteries and charging networks).

The role of the Public Administration is clearly critical as far as environmental regulations that indirectly promote the use of electric vehicles are concerned. In the case of Europe these include: 1) Directive 2009/28/EC which states that 10% of the energy used in transport must be provided by renewable sources by 2020. 2) The EC regulation 443/2009 which imposes reductions in average emission levels for vehicle manufacturers, setting objectives of 130 gCO2/km for 2015 and 95 gCO2/km for 2020. 3) The European strategy to promote the use of environmentally friendly vehicles (COM, 2010; 186 final) which establishes as priorities the development of electric vehicles that are at least as safe as conventional ones, a European standard for charging points, a public charging network, a smart grid and research programs for the safe recycling of batteries.

If we examine a number of pilot projects implemented in various cities around the world, we can see how the nature and extent of public intervention have changed considerably. Wiederer and Philip (2010) present case studies in four cities that have

<sup>&</sup>lt;sup>7</sup> For a review of energy efficiency in the car industry see Jiménez et al (2012b).

run pilot schemes for the introduction of electric vehicles and which provide examples of the roles that the public sector might adopt.

- Singapore, in June 2010, initiated a project to invest 20 million dollars in setting up a comprehensive network of recharging points, and to provide subsidies for the purchase of electric vehicles. The primary goal of local government is to attract the electric car industry to Singapore.

- The Indian city of Bangalore has no specific plan to promote electric vehicles, yet there are over a thousand electric vehicles of the REVA brand (a domestic producer) on the streets. This seems to indicate that, at least in this particular case, there is no need for active intervention on the part of the public authorities to promote demand for electric vehicles.

- In the City of London, the scheme has entailed a 17-million pound investment, including the installation of a network of charging stations, the electrification of its public transport fleet and incentives for purchasing and marketing electric vehicles. To develop this ambitious project (submitted in May 2009), the city council's transit agency "Transport for London" is working with a consortium of electric vehicle manufacturers, major utilities in London and car rental companies.

- The City of Berlin has initiated two simultaneous electric vehicle programs, both powered and funded by private industry. The city is administratively limited to helping companies and ensuring the compatibility of the two offers as far as the charging network is concerned.

It is interesting to note how the pilot projects run in these cities have given different emphases to the deployment of electric vehicles: industrialization, in the case of Singapore; full network development, in the case of London; and the recharging network, in the case of Berlin. Likewise, the degree of public sector involvement varies significantly from one project to another: from simple guidelines for private companies (the case of Bangalore and Berlin) to an active role in the market though heavy investment (the case of Singapore and London). The sections that follow break down the measures applied by the public authorities as they seek to address the main barriers and to promote the development of electric vehicles.

#### 3. **Recharge system**

Before examining in greater depth the possible barriers that the recharge system might represent to the introduction of electric vehicles, and the main measures adopted by the public sector in relation to them, we outline the various types of recharge and their main characteristics.

As Wiederer and Philip (2010) report there are three types of system: first, slow recharging points or Level 1 points, located primarily in homes, apartment buildings or in public spaces close to residences. It is thought that car owners will use these stations essentially to recharge their vehicles over night, that is, when electricity consumption is in the valley period. However, workplace parking lots or shopping centres are also potential sites for these points and consumers are, therefore, more likely to recharge their vehicles during the day, that is, during peak periods of electricity demand. This may necessitate an increase in the capacity of electricity generation.

Second, there are the rapid charging or Level 2 points, located primarily in shopping centres, supermarket car parks or gyms, and which will also be used during the day. And third, the super-fast or Level 3 charging points, which will be located in line with existing service stations along the highways. The following table summarises the different types of charging points and their main features.

Table 1: Type of recharge systems			
Charging level	Specification	Typical use	Time to charge battery
Level 1 (slow)	120V / 13 A	Charging at home / office	7- 8 hours
Level 2 (Fast)	240V / 32 A	Charging at supermarket , gym	3 – 4 hours
Level 3 (Rapid)	Up to 500V / 200 A	Like a normal gas station	30 minutes

- - -

Source: Wiederer and Philip (2010)

As indicated in the International Energy Agency report (2008) slow recharge systems are currently almost non-existent, with the exception of a few pilot programs and schemes. In the case of rapid charge systems, the report identifies the need for the deployment of such points in conjunction with the development of battery "swapping" systems. Level 2 points should, it recommends, be located primarily in locations of high-density traffic, stores or public car parks.

High battery costs would make the "swapping" system a viable alternative. However, it would require compatibility between models of electric vehicles and their battery systems. But Wiederer and Philip (2010) believe it unlikely that such standardization can be achieved, so that a "swapping" system would not be able to replace Level 2 points. Yet, such systems might be an alternative in very specific cases such as for a fleet of city taxis where a uniformity of vehicle type (and, hence, of battery type) could make "swapping" a competitive possibility.

Having defined the different types of recharge on the market, we now examine the features and elements that need to be taken into account when developing an efficient charging network.

The first point to bear in mind is the diffusion of information to drivers regarding the location of these charging points (International Energy Agency, 2008). GPS technology can be useful for informing consumers about the nearest available points and about free parking places where vehicles can be recharged. Such information would reduce driver uncertainty and stress, enhancing the utility of electric vehicles and reducing the number of optimal charging points.

A second issue raised by the aforementioned report is the need for standardised charging systems, at least on the same continent. The main features that would need to be standardized are: 1) Plug-in types; 2) Recharging protocols; 3) Communication protocols between cars and recharging systems; 4) Regulations for public recharging that ensure safety with minimal administrative barriers; 5) Battery recycling standards and regulations; and 6) Utility regulations conducted by state/provincial authorities to ensure orderly participation in this market.

A third element to consider is the cost of these charging points, and the tariff system for recovering that investment. The International Energy Agency report (2008) estimates the cost of such infrastructure at between 1,000 and 2,000 dollars per vehicle.

Wiederer and Philip (2010) provide more detailed information by estimating the costs of recharge points by type. For Level 2 points designed for private use (located in private homes or garages), the cost ranges between 500 and 2,500 dollars. If the point of recharge (Level 2) is publicly accessible (located in public garages or on the street), the cost rises to between 2,000 and 8,000 dollars. Finally, for Level 3 points (located along highways and requiring a maximum of 30 minutes to recharge a vehicle) the cost ranges from 40,000 to 75,000 dollars.

Based on these estimated costs, the study analyzes the price (under different levels of use) that would make investment in a charging network viable. According to the authors, the price of electricity supplied to the public charging points (especially at Level 3) would have to be raised considerably - more than 70% on current levels rising to as much as 238%. Such tariff rises would make public charging points quite unattractive to potential consumers of electric vehicles, who logically would opt to recharge their cars at home at greatly reduced rates. Thus, the deployment of a network of public recharging points financed by private companies might be rendered impractical, especially in the case of Level 3 points.

To date, governments and industry have yet to reach an agreement on who should cover these investment costs and how they might be recovered through the charging of fees. It is clear that a major investment in charging infrastructure is required and that national, regional and local authorities will have a significant role to play. The International Energy Agency report (2008) itemises a set of measures that might facilitate the optimal development of charging networks:

- Analyze each region to estimate the relation between demand from electric vehicles and electricity supplies, especially after the initial phase when the demand for electric vehicles is predicted to grow.

- Establish appropriate codes and standards for recharging power supplies and for smart metering.

- Analyse development strategies at the national level to identify infrastructure and priority areas, schedules and funding.

- Define the roles and responsibilities of various stakeholders (governments, regulators, utilities, vehicle manufacturers and consumers) to establish a clear strategy of collaboration and cooperation between different levels of government, as well as with companies and electric vehicle manufacturers.

- Prioritize domestic recharge without neglecting the implementation of a development plan for commercial charging stations, especially as the number of electric vehicles increases.

- Explore the feasibility of various approaches to fast charging methods, such as the "swapping" system.

A fourth point to consider is the role to be played by the smart grid, including the need for next-generation infrastructures and future technological developments. Analyses should also be undertaken of consumer availability for electricity sales and the circumstances under which this might occur.

These four points need to be given careful consideration in the development of an efficient charging network and in ensuring that the charging network does not become a barrier to the growth of the electric vehicle market (Kley et al, 2011). However, these authors point out that with a charging network in private homes, the public charging network may be minimal. Furthermore, they evaluate the problem of financing public charging points, given the low percentage use that they might attract.

Kley at al (2011) likewise identify a set of barriers that might limit or prevent the development of charging networks. They identify three types of barrier: 1) regulatory barriers, particularly the need to fix standards regarding recharging and conditions affecting the sale of electricity (to allow power companies to design their own investment plans) and permits for the installation of charging points; 2) economic barriers, including the uncertainty regarding user demand for recharge, as well as the type of fees to be paid by users at charging points; 3) technological barriers, principally the uncertainty of recharge technology and future applications of smart grids.

The International Energy Agency report and RETD (2010), among others, have outlined joint plans for the introduction of electric vehicles and charging networks in different countries. These are summarized in the following table.

Country/Location	OEM	Deployment plan	Timeline
France: Strasbourg	Toyota/EDF	100 vehicles (Prius PHEV)	2009-2013
France: Paris	Toyota/EDF	4,000 vehicles (autolib)	2011
		1,400 charging stations	
Germany: Berlin	Daimler/RWE	100 vehicles (Smart EV)	2009
		3,600 charging stations	
Japan: Tokyo	Mitsubishi,	200 charging stations	2009
	Subaru/Tepco		
Spain: Sevilla,	Various	2,000 vehicles	2009-2011
Barcelona, Madrid		550 charging stations	
Denmark	Better Place	500,000 charging stations	2011
	Gov.	50 EV's charging stations in 2009	
		150 EV/ charging stations in 2012	
Ireland		10% of fleet electric (250,000 veh)	2020
		in 2020	
		1 M€ R&D and demo plan	
Israel	Better Place	100,000 charging stations	2010
Italy: Rome, Pisa	Daimler/Enel	100 vehicles (Smart EV)	2009-2013
		400 charging stations	
Netherlands:		10,000 vehicles to 40,000	2015-2020
Amsterdam		vehicles	
Canada: Vancouver	Mitsubishi	Unknown (i-MiEV)	2009
USA: Arizona,	Nissan	Up to 1,000 vehicles (Nissan	2010
California, Oregon,		Leaf)	
Tennessee,		12,750 charging stations	
Washington			

Source: IEA-RETD (2010)

Below we describe in greater detail some of these initiatives by broad geographical region. First we examine the development of the charging network in North America (United States and Canada), then in Europe, and conclude by analysing initiatives in the rest of the world.

#### **North America**

The Boston Consulting Group (2009) estimates the amount of investment needed to develop the public charging network in the United States at 21 billion dollars. The participation of energy companies, particularly utilities, in the development of this public network will, it concludes, be far from straightforward. As discussed above, a public charging network would generate only limited revenue growth, while the need for investment is high and the risks substantial. The increase in demand for electricity that

would be created by the mass introduction of electric vehicles would not have a significant impact on the earnings of these companies; however, they would have to invest heavily in a public charging network near residential areas that might not be attractive to consumers. Note that the electricity prices charged as a result of these investments would be more than twice those charged to home residents, which could well mean that the majority of consumers would not contemplate using these public recharging points. Thus, the participation of power companies in the development of public recharging systems seems unlikely without strong public incentives.

KPMG (2010) reports that two separate bills were presented before the Senate and the House in May 2010 to increase funding for electric vehicle infrastructure: the Senate bill earmarks up to 250 million dollars for 15 municipalities and cities to improve their infrastructure. The House bill offers up to 800 million dollars to five municipalities and 2,000 dollars to electric vehicle owners to install charging equipment.

To fund the charging system and to ensure their economic viability, both Kley et al (2011) and Wiederer and Philip (2010) suggest that annual fees may be a more efficient manner of generating the necessary resources than user payments. Wiederer and Philip (2010) estimate this annual payment at between 100 and 200 dollars, depending on the number of charging and recharging points. Whatever type of business plan is eventually chosen to fund the charging system it should serve eventually to reduce owner "anxiety", but doubts remain regarding their economic viability.

#### Europe

Within the European market we report here on three initiatives: one local and two regional.

At the local level, the City of London plans to establish a network of recharging stations throughout the metropolitan area. The goal is to set up 25,000 recharging points by 2015, including a fast-charging network, to be distributed as follows: 500 in the street, 2,000 in car parks and 22,500 in collaboration with the business community and to be located in work places and points of entertainment (Mayor of London, 2009).

In Spain, the Generalitat de Catalunya has devised an ambitious project to deploy 91,200 charging points by 2015. The plan includes the creation of both private and

15

public charging points. The following table shows the specific objectives over the forthcoming years.

Table 3: Objectives of charging stations in Catalonia						
Type of charge	Number of charging stations					
	2010	2011	2012	2013	2014	2015
Private	600	4100	13000	26000	44800	76000
Private of reinforcement	60	410	1300	2600	4480	7600
Public (park station)	90	230	1040	2080	3584	6080
Public (street)	40	70	130	520	896	1520
Total	790	4810	15470	31200	53760	91200

Source: Generalitat de Catalunya (2010)

As can be seen from the above table, the Generalitat de Catalunya seeks to establish private recharging points for each electric vehicle, while creating a network of public charging points to increase the project's credibility in the eyes of vehicle owners and to reduce their "range anxiety".

To achieve these objectives, the Catalan Government has proposed a set of public support programs and incentives. In the case of private charging points, public support is focused on two features: 1) providing a line of credit to owners of electric vehicles for the installation of private charging points (provided recharge is conducted in valley hours) involving minimum technical requirements as regards legalization; and 2) promoting the installation of charging points in new buildings, especially in public housing developments. In the case of the public charging network, the government is offering subsidies of up to 30% of the investment (up to 600,000 euros) if a private company undertakes to provide the service and grants of up to 40% of the investment for projects aimed at supra-municipal electric mobility (up to a maximum of 200,000 euros). Similarly, the Generalitat de Catalunya aims to reach agreements with the operators of parking lots and municipal entities to ensure that 1.5% of all such lots have a charging point by 2015.

Finally, Meier-Eisenmann et al (2001) report the behavioural patterns displayed by electric vehicles and their owners, by employing a range of methodologies (monitoring charging systems, surveys conducted directly with owners, etc), in their analysis of a pilot scheme conducted in the Swiss canton of Ticino. They report that 56% of owners of electric vehicles use the parking areas reserved specifically for them, rising to 82 percent in the case of four-wheeled vehicles (two-wheeled vehicle owners use these spaces much less frequently - 29%).

Most electric vehicle owners, therefore, take advantage of the reserved parking spaces for recharging their batteries. Specifically, 68 percent claimed to "always" or "almost always" park their vehicles in these reserved spaces. Thus, public charging points located in parking spaces reserved specifically for electric vehicles are popular with owners, above all those of four-wheeled vehicles.

#### Rest of the world

Outside North America and Europe mention should be made of the Electric Recharge Grid Operator model, and various initiatives taken in Japan, a pioneer in this sector.

Andersen et al (2009) report that Israel was the first country to adopt the Electric Recharge Grid Operator<sup>8</sup>, introducing the first prototype electric vehicle in Tel Aviv in May 2008. Plans to develop the network have been underway since 2009. Denmark was the second country to adopt the model in a joint project between the government and country's main electric utility company, DONG Energy. Later adopters include Australia (in a joint project involving the State of Victoria, the AGL electricity company and the financier, Macquarie Bank) and various municipalities in the Bay Area of California and Hawaii.

Japan boasts over two decades of experience in this sector (see Ahman, 2006, for a summary of all public policies for the development of electric vehicles in this country).<sup>9</sup> In 1993 Japan initiated a project (ECO-Station Project) aimed at introducing 2,000 recharging stations for clean energy vehicles by 2000. Of that number, around half were for electric vehicles (Hayashi et al, 1994). The KPMG report (2010) notes that in March 2010 Toyota, Nissan, Mitsubishi, Fuji Heavy Industries and Tokyo Electric

<sup>&</sup>lt;sup>8</sup> The Electric Recharge Grid Operator business model creates a market for the coordinated production and consumption of renewable energy (see Andersen et al, 2009).

<sup>&</sup>lt;sup>9</sup> Limited not solely to electric vehicles, but extended to other projects aimed at reducing pollution, such as hydrogen vehicles. The WE-NET program of NEDO considers this and the necessary infrastructure.

Power Company established the association CHAdeMo (Charge Move), in order to install standardized, rapid charging points and equipment worldwide.

#### 4. Stimulating demand for electric vehicles

Electric vehicles and plug-in hybrid vehicles can generate a set of significant environmental and cost advantages over their entire useful life cycle, greater than those afforded by traditional ICE vehicles. However, as the Boston Consulting Group report (2009) points out, consumers are most heavily swayed by purchasing price when making consumer decisions, and electric vehicles continue to be more expensive in this regard. Kageson (2005) reports that hybrid vehicles are between 20 and 30 percent more expensive than ICE vehicles.

According to this same report, electric vehicles will remain unattractive unless the acquisition cost is subsidized.<sup>10</sup> The high price of these vehicles, it is claimed, is primarily attributable to the cost of the battery, estimated at \$700 per kWh. And only if this cost can be reduced to \$500 per kWh, with oil prices remaining at levels between \$100 and 120, can electric cars begin to compete with ICE vehicles. The report concludes that public subsidies are essential, therefore, to stimulate demand in the sector. Examples of such subsidies include the 7,000-euro grant offered by the French government for the purchase of electric vehicles, and similar measures adopted in Denmark and Israel.

In addition to such subsidies, or any other type of intervention aimed at reducing the purchasing price, a further measure identified for promoting electric vehicles is the raising of pollution standards for ICE vehicles. As reported in the Transport & Environment document (2009a), fixing contamination levels for ICE vehicles at 80 g CO2/km by 2020 and at 60 g CO2/km by 2025, accompanied by increases in gasoline taxes, would result in a competitive upgrading for electric vehicles, thus increasing market penetration.

Transport & Environment (2009c) reports, however, the potential dangers of considering electric vehicles as zero emission generators. This would be to ignore, first

<sup>&</sup>lt;sup>10</sup> Various countries have subsidized the price of electric vehicles to ensure a minimum demand, especially in the initial stages of their introduction. The International Energy Agency report (2008) sets out the objectives Governments seek to achieve in this regard.

and foremost, the emissions created in the generating of the electricity consumed by the vehicles. Moreover, as the deployment of electric vehicles would statistically reduce average vehicle emissions, manufacturers would be able to justify selling more ICE vehicles while continuing to comply with current legislation on average emission levels.

Taxes represent a further instrument that could be used to promote demand for electric vehicles. Raising the taxes on gasoline consumption or reducing those on electricity (produced primarily from renewable sources) would modify the relative prices of energy products, favouring the introduction of electric vehicles (see Andersen et al, 2009).

A final option for stimulating demand is to promote the use of public vehicle fleets, courier companies and the like. Kley et al (2011) report that various delivery companies have introduced electric vehicles into their fleets. UPS has started to use electric vehicles for postal services in the Washington area. Similar programs are also being introduced in Europe, where companies such as DHL or TNT are using electric vehicles in cities such as Hannover, Barcelona and Lyon.

Below we outline the main measures being taken to foster electric vehicle demand around the world.

#### **North America**

The US Department of Energy (2011) aims to put 1 million electric vehicles on the country's roads by 2015. To achieve this goal, the US has implemented a series of measures to promote the demand for such vehicles. Gallagher and Muehlegger (2011) report that these actions are being performed locally, as well as at state and federal levels. The following table summarises the various initiatives being undertaken by a number of States and cities in the US.

			,			
Single –	Income	Sales tax	Vehicle	State gov.	Registration	Parking fee
occupancy	tax	exemption	emissions	Purchasing	or excise	reduction or
HOV lane	credit		test	requirement	tax	exemption (cities)
access			exemption		exemption	
AZ (pilot)	CO	CT+	CO+	MN	DC	Alburquerque,NM
CA+	MD*	DC	MD	NM	IL+	Austin,TX
CO (on hold)	NY+*	ME*	WA	NY	NM	Baltimore,MD
FL	OR	NM+		WI	PA	Ferndale,MI
GA (on hold)	PA					Huntington,NY
NJ	SC					Los Angeles,CA
NY (pilot)	UT*					New Haven,CT
UT	WV*					Salt Lake City,UT
VA						San Antonio,TX
						San Jose,CA
						Santa Monica,CA
						Vail,CO
						Westchester,NY

Table 4: State subsid	y for h	ybrid	vehicles	in	United	States

Note: + denotes incentive targeted at high fuel-economy hybrid vehicles, and \* denotes expired program.

AZ (Arizona); CA (California); CO (Colorado); CT (Connecticut); DC (Federal District of Washington DC); FL (Florida); GA(Georgia); IL (Illinois); MD (Maryland); ME (Maine); MI (Michigan); MN (Minnesota); NJ (New Jersey); NM (New Mexico); NY (New York); OR (Oregon); PA (Pennsylvania); SC (South Carolina); TX (Texas); UT (Utah); VA (Virginia); WA (Washington); WI (Wisconsin); WV (West Virgina).

Source: Gallagher and Muehlegger (2011).

Loans and tax exemptions are the most popular measures introduced to date to promote hybrid vehicles (along with such initiatives as access to HOV lanes, and reductions in the price of public car parks, which we discuss below). Gallagher and Muehlegger (2011) provide a detailed summary of these support programs in the US. At the federal level support for the purchase of hybrid vehicles took the shape of a tax break of \$2,000 between 2000 and 2005. From January 2006 onward, this became a vehicle loan, the amount of which is linked to the model. In general, these credits are greater than the earlier tax deductions and range from \$3,150 for a Toyota Prius to \$650 for a Honda Accord Hybrid and the Saturn VUE Green Line. These credits are limited to the first 60,000 units sold by each manufacturer. Toyota and Honda actually exceeded this limit in May 2006 and August 2007, respectively.

State wide, there is considerable variety in the amount of aid provided for the purchase of hybrid vehicles. The following table reports the amounts provided by individual States.<sup>11</sup>

<sup>&</sup>lt;sup>11</sup> Busse et al (2006) show how the characteristics of car price promotions can affect consumers. For conventional vehicles, sales from producers direct to consumers can result in a 70 to 90 percent discount in the final price, while sales from car dealers reduced final prices for consumers by around 40 percent.

State	- D	Madala accord	Generosity	
Income tax credits	Duration	Models covered	range	
Colorado	2001 – present	All, but VUE, GS450h, and Camry*	\$2265-\$6542	
Maryland	2001 – 2004	Civic, Prius, Insight	\$1000	
New York	2000 – 2006	All	\$2000	
Oregon	2003 – present	All	\$750-\$1500	
Pennsylvania	2006 – present	Civic, Prius, Insight, Escape	\$500	
South Carolina	2006 - present	All	\$130-\$630	
Utah	2001 – 2005	Civic	\$1537-\$1720	
West Virginia	2003 - 2006	All	\$2411-\$3750	
Sales tax Waivers				
Connecticut	2004 – present	Civic, Prius, Insight	\$1217-\$1409	
District of Columbia	2005 – present	All	\$1226-\$3294	
Maine	2000 – 2005	Civic, Prius, Insight	\$300-\$500	
New Mexico	2004 - present	Civic, Prius, Insight	\$608-\$704	

#### Table 5: State tax incentives in United States

\*Colorado income tax credits for the VUE, GS450h and Camry begin post-2006. Generosity for Sales Tax Waivers in CT, DC and NM are estimated based on vehicles MSRP.

Source: Gallagher and Muehlegger (2011)

Beresteanu and Li (2011) analyze the effectiveness of such public support plans for increasing the demand for hybrid electric vehicles. They show how these programs have increased sales of hybrid vehicles by 20 percent. The report also discusses the effect on sales of reducing the cost of a barrel of oil to 75 dollars and cutting CO2 emission costs to \$177 per ton. The study conducts a simulation to show that the sales of hybrid vehicles would have been between 21 and 38 percent lower if gas prices had not risen between 2001 and 2006. Thus, it appears that the evolution of gasoline prices has a greater impact on the demand for hybrid vehicles than does public support.<sup>12</sup>

<sup>&</sup>lt;sup>12</sup> Alberini et al (1995) show, using a theoretical model and applying it to the Delaware State program, that the percentage of consumers who changed their vehicle thanks to the aid program was around 5%. Huang (2010) analyses the "cash for clunkers" program in the US. He concludes that an average grant of \$4,200 is able to convince between 25 and 30 percent of consumers to change their vehicle for a more energy efficient one. Sivak and Schoettle (2009) note that the energy efficiency of vehicles purchased improved between 0.6 and 0.7 miles per gallon between July and August 2009 thanks to this same program. For a detailed description of the "cash for clunkers" program and other programs for the replacement of older vehicles with more energy efficient ones see Yacobucci and Canis (2010) or Cooper et al (2010).

Diamond (2009) similarly analyses the effectiveness of public aid plans to promote hybrid vehicles. The author concludes that federal aid is of limited effectiveness, resulting in an increase in demand for hybrid vehicles of 18 percent.

It seems such aid programs could have a major impact on reducing emissions, if support was not geographically uniform. Skerlos and Winebrake (2010) claim that if the aid targeted consumers in densely populated metropolitan areas (where contamination is greatest), those in areas with a cleaner power technology mix and those with lower income levels, the adoption of electric vehicles would be higher and it would have a greater impact on pollution levels.

Tax relief programs to increase demand for hybrid vehicles have also been launched in Canada. Chandra et al (2010) report that 26% of hybrid vehicles sold in that country can be attributed to public grants. This increase in hybrid vehicle sales represents a reduction in the sales of mid-range cars, SUVs and some high-performance compact vehicles. The substitution is explained by the similarity in price and characteristics presented by the hybrid vehicles sold in Canada (mainly the Toyota Prius and Toyota Camry).

The authors note that purchases of these hybrid vehicles mean that the average cost of reducing one tonne of CO2 now stands at \$195. The aid program was implemented in 2000 when the province of British Columbia introduced a tax exemption for the purchase of hybrid vehicles. In 2006, a further five provinces implemented similar policies. In 2007, the Federal Government started a two-year program to promote the sale of energy-efficient vehicles or vehicles that use alternative fuels through cash rebates.

In 2010 the Government of Ontario offered grants of between \$4,895 and 8,321 for the first 10,000 customers who purchased a new electric or hybrid vehicle. The following table (Table 6) shows the main initiatives taken and their characteristics.

Province	Vehicle eligibility	Rebates and timing
British Columbia Policy announced: August	All hybrid vehicles with regenerative braking (cars	30% of tax paid up to maximum of \$500 for vehicles
PST rate: 7% with graduated	and SUV's eligible)	30% of Provincial Sales Tax
increases for vehicles over \$55K		(PST) paid up to maximum of \$1000 after July 31st 2001 A point of sale reduction of all PST to a maximum of \$2000 after February 16th 2005 Additional rebates in PST (reductions in the graduated increase of PST over 7%) for
		nybrid venicles over 55K (see note 1)
Prince Edward Island Policy announced: March 2004 PST rate: 10%	All hybrid vehicles are eligible	All the PST paid up to \$3000, for vehicles bought after March 30th 2004
Ontario Policy announced: May 2001 PST rate: 8%	All hybrid passenger cars (with regenerative braking) eligible 2001, SUV's eligible 2002	PST rebate up to a maximum of \$1000 for cars bought after May 10th 2001 Hybrid SUVs and trucks included June 18th, 2002 A point of sale reduction of all PST to a maximum of \$2000 after March 23rd 2006
Quebec Policy announced: March 2000 PST: 7.875%	See notes (2) below	All PST paid to a maximum of \$1000 for vehicles bought after March 23rd 2006 and before February 21st 2007 All PST paid to a maximum of \$2000 for vehicles bought after February 22nd 2007 and before January 1st 2009
Manitoba Policy announced: November 15th 2006 PST rate: 7%	See notes (3) below	Flat \$2000 rebate for all vehicles bought after November 15th 2006

able 6: Measures to	promote	demand i	n Canada
---------------------	---------	----------	----------

Source: Chandra et al (2010)

Notes: People buying light vehicles that are priced >\$55,000 have to pay a higher PST rate. This rate increases by 1% for the first \$1000 over \$55,000 and continues to increase by 1% for every additional \$1000 to a maximum of 10% (for vehicles costing more than \$57,000). For hybrid vehicles the graduated increases come with an additional exemption of \$7,000 on the threshold. This means that the PST does not increase for hybrid vehicle until their price reaches \$62,000.

Cars eligible for a rebate in Quebec are: 2005 and 2006 Honda Insight; 2005–2007 Toyota Prius; 2007 Toyota Camry Hybrid; 2008 Ford Escape Hybrid (two-wheel drive); 2005–2007 Honda Civic Hybrid; 2005 Honda Accord Hybrid; 2007 Nissan Altima Hybrid.

Cars eligible for a rebate in Manitoba are: Honda Insight; Lexus GS 450H; Lexus RX 400H; Toyota Camry Hybrid; Toyota Highlander Hybrid; Toyota Prius; Chevrolet–Silverado 1500 LS Hybrid; Ford Escape Hybrid; GMC Sierra 1500 SLE Hybrid; Honda Accord Hybrid; Honda Civic Hybrid; Saturn VUE Green Line.

Gallagher and Muehlegger (2011) note that it is not only the Public Administration that offers grants, but that private corporations have begun to provide incentives to employees to purchase hybrid vehicles. In 2004, Timberland returned \$3,000 to employees who bought a hybrid vehicle. Google offered \$5,000 from March 2005, while the Bank of America returned \$3,000 to its employees from June 2006.

#### Europe

As discussed in section 2, the European Union has introduced regulations to promote the introduction of electric vehicles in its general stock of cars. Denmark has also implemented measures at the national level to promote demand for electric vehicles. It has used the tax system to grant full exemption from taxes on electric vehicles until 2012, making a budget provision for 2008-2011 of more than 35 million DKK.

At the regional level, the Spanish Autonomous Community of Catalonia (Generalitat de Catalunya, 2010) has introduced various measures to promote demand for electric vehicles, aiming for a target of 76,000 vehicles by 2015. One of the key measures here is the introduction of electric vehicles in the Government's own fleet of vehicles. The document seeks to ensure that 15% of public service vehicles are electric by that year. The second of the measures implemented in Catalonia is a system of grants for such vehicles. During 2011 the Generalitat de Catalunya provided the following grants depending on the type of vehicle: 1) for passenger cars, a contribution up to  $\in$  7,000 to a maximum of 15% of the cost of the vehicle; 2) for electric motorcycles (power 4kWh),  $\notin$  750 euros to a maximum of 15% of the cost of the cost of the vehicle; 3) for commercial vehicles, up to  $\notin$  50,000 to a maximum of 15% of the cost of the vehicle. This amounted to an expenditure budget of 1.5 billion euros in 2009.

The report to promote electric vehicles in Catalonia stresses the need to provide more and better information to users, so as to overcome the barriers discussed above in Section 2, namely the obstacles to innovative change. The report concludes that the Public Administration need to play a proactive role in providing information in all areas related to electric vehicles: purchase decisions, maintenance, life cycles, associated technologies, etc.

Finally, mention should be made of two local initiatives taken in London and Amsterdam. In the case of London, the Mayor offers discounts on the city's congestion charges (worth up to 1,700 pounds a year), and encourages the uptake of electric

vehicles via car clubs (Mayor of London, 2009). The British government also offers buyers of electric vehicles a 5,000 pound subsidy (KPMG, 2010). In the case of Amsterdam, the City Council Electric Transport subsidy program has a fund of EUR 3 million and covers up to 50 per cent of the additional costs incurred in buying an electric vehicle (KMPG, 2010).

#### Rest of the world

In the rest of the world, the first initiative to highlight is that taken in Israel. Here, the government amended its tax system in January 2008, halving the taxes paid on electric vehicles compared to rates for ICE vehicles. Israel classifies its vehicles into 15 categories based on a "green index" (on a scale from 0 to 100 "green points"), where the greener the car, the lower is the tax rate to be paid.

China has also taken measures to increase demand for electric vehicles. Weinert et al (2008) report that the Chinese government has promoted demand for electric motorcycles (E2Ws) by introducing changes in the local regulations of various municipalities (beginning in just 30 cities in 1998 this has now expanded to 148 cities in 2006) so as to reduce ICE motorcycles.

The International Energy Agency (2008) reports that China has also established programs to promote the electrification of vehicles nationwide. From 2008, the Ministry of Science and Technology initiated a series of projects to introduce vehicles using alternative energy in 10 cities. The aim was to reach a total of 500 electric vehicles by 2009 and around 10,000 units by the end of 2010. To this end, subsidies for the purchase of electric vehicles of 50,000 RMB (about \$7,300) were introduced, although only the F3DM model benefits from this grant.

Since 1978 Japan has promoted several leasing programs and other incentives to encourage the purchase of cleaner electric vehicles (Iguchi, 1992). Ahman (2006) describes how, under the Environment Conservation Programme in 1995, the Japanese government announced the replacement of 10% of its public vehicles by 2000 with vehicles producing lower emissions. In 2001, the government also established a goal to replace all used vehicles with cleaner alternatives, of which 60% were expected to be hydrogen-electric mix (EVAAP, 2002). In addition, in the 90s various policy measures were adopted aimed at increasing demand for cleaner vehicles (methanol-fuelled vehicles and CNG vehicles).

The city of Delhi (India) has also implemented an aid program (15% discount) for the purchase of electric vehicles. In states such as Madhya Pradesh, Kerala, Gujarat and West Bengal, the excise tax on the purchase of electric vehicles has been reduced by up to 4%, while other cities and states in India have implemented reductions in road tax or vehicle registration taxes.

#### 5. Industrialization and R&D policies

As mentioned throughout the paper, battery costs are the key factor that will determine whether electric vehicles can ultimately be competitive and win market share at the expense of conventional ICE vehicles (Offer et al, 2010). Little wonder, therefore, that most R&D programs target technological improvements to vehicle batteries. The cost of electric vehicle batteries is dictated primarily by their size, weight and energy density, and any progress in these areas would result in substantial improvements to the competitiveness of these vehicles.

The International Energy Agency report (2008) estimates that between 2012 and 2015 lithium batteries will cost between 300 and 600 dollars per kWh. This means that for a battery with a capacity of 20 kWh (the minimum for an electric vehicle), the cost would rise to between \$6,000 and \$12,000. Clearly, the nearer this price can be kept to the lower band range the better for the viability and competitiveness of electric vehicles. Economies of scale and a decrease in the learning curve should help achieve these cost reductions, but they cannot be guaranteed.

Given the importance of the battery in the total cost of electric vehicles it is not surprising that producers of these vehicles have initiated R&D projects in conjunction with battery producers aimed at improving production of this key element. The following table summarizes some of the projects initiated by vehicle and battery manufacturers.

Car manufacturer	Battery manufacturer
BYD Auto	BYD Group
Fiat-Chrysler	A123 Systems, Altairnano
Ford	Johnson Controls-Saft
GEM	Sanyo / Panasonic
GM	LG Chem
Hyundai	LG Chem, SK Energy and SB LiMotive
Magna Group	Magna Steyr
Mercedes-Benz	Continental, Johnson Controls-Saft
Mitsubishi	GS Yuasa Corporation
Nissan	AESC
REVA	Indocel Technologies
Renault	AESC
Subaru	AESC
Tata	Electrovaya
Think	A123 System, Enerdel / Ener1
Toyota	Panasonic, EV Energy
Volkswagen	Volkswagen and Toshiba Corporation

Table 7: Joint R&D programs initiated by car and battery manufactures

Source: International Energy Agency (2008)

The International Energy Agency report (2008) recommends that governments provide backing for battery producers, especially the most innovative in the sector. Such support should enable the construction and expansion of battery production plants ensuring that investment requirements are not an obstacle to progress. Access to materials such as lithium in the medium and long term makes it increasingly necessary to develop R&D programs that promote the use of materials and innovative designs that can reduce the production costs of electric vehicles. Production costs can be further reduced if battery producers and electric vehicle manufacturers can operate together.

The International Energy Agency report (2008) also stresses the need for technology development programs for the recycling of the batteries from electric vehicles once their life-cycle has ended.

Below, we describe the main initiatives taken by Governments around the world.

#### North America

The US authorities have been granting support to the electric vehicle industry for many years. As indicated by the Transport and Environment report (2009a), as early as 1973 the US Congress passed the "Electric and Hybrid Vehicle Research, Development, and Demonstration Act" designed to promote new technologies related to electric vehicle

batteries and the engines of such vehicles. Despite its ambitious nature, the plan failed to achieve its objectives and was shelved by the Reagan administration.

As part of the more recent US strategic plan (US Department of Energy, 2011), the "American Recovery and Reinvestment Act" (ARRA) greatly expands the resources available for industrial investment, a significant portion of which are assigned to the energy industry. These resources include "tax credits" to build vehicles powered by batteries. Indeed, the US has set itself the goal of being able to produce 500,000 hybrid electric vehicles in 2015.

The US Department of Energy (2011) has encouraged investments through its Office of Science in basic research, channelled through the Massachusetts Institute of Technology (MIT). This has resulted in the development of a new material known as "new nanostructured cathode" which can be applied to battery technology. A company, A123 Systems (Watertown), has been set up to commercialize this new technology and it has been the recipient of aid from the "Small Business Innovation Research Department" since 2002 and the "Office of Energy Efficiency and Renewable Energy" since 2006.

Other lines of support for the development of R&D include the "Partnership for a New Generation of Vehicles (PNGV)" and its successor the "FreedomCAR", as well as specific programs for electric vehicles. According to Gallagher and Muehlegger (2011), these programs account for a total expenditure of \$80 million from the US budget.

#### Europe

The European Commission has also funded R&D projects via their programs JOULE I and II (Transport & Environment, 2009a). Among the entities receiving financial support are "European Electric Road Vehicle Association" (AVERE) and the "Association of Cities Interested in the Use of Electric Vehicles" (CITELEC).

A number of countries have also undertaken their own programs of technological development, working from different perspectives. Thus in France, Italy and Sweden several companies are jointly conducting R&D programs in the development of electric vehicles, albeit under the auspices of different governments. In Germany, however, it is the private sector that is making these investments, while the Government has taken a much more laissez faire approach.

In the case of Spain, the Catalan administration has not neglected R&D policies in its overall strategy for the introduction of electric vehicles. It two main policy proposals to promote industrialization and R&D in the sector include 1) an attempt to establish a battery-pack manufacturer in Catalonia; and 2) communication projects between the grid and electric vehicles budgeted to the tune of 35 million euros for the years 2011 and 2012. Other initiatives have also been taken (with a budget of 173 million euros) to promote research projects related to the development of electric vehicles.

#### Rest of the world

Japan's industrial policy in relation to the development of electric vehicles (including vehicles powered by clean energy) has been highly active, as evidenced by Ahman (2006). The MITI has promoted electric vehicles (BPEVs) since 1971, launching a five-year government-industry R&D programme. The MITI has also funded company R&D programs. Between 1978 and 1996 it lent support to leasing projects (MITI, 1990). The Japanese government, through MITI, has promoted various policies for "Demonstrating the Feasibility of BPEVs" in combination with Intelligent Transportation Systems, research on lithium batteries, and developing high-energy efficient hybrid vehicles (see Ahman, 2006).

Initiatives in China should also be highlighted. As Brown et al. (2010) report, this country has allocated ten million Yuan (1.46 billion dollars) to a program to help the auto industry carry out technological innovation projects.

#### 6. **Program development and management of sustainable road mobility**

In this section we draw on the proposals contained within the general plan to promote electric vehicles, as drafted by the Generalitat de Catalunya (Catalonia, Spain), in order to explain the measures that can help introduce such vehicles within programs of sustainable mobility. The following specific measures are forwarded:

1) A reduction in the time and financial costs of travel. Permission for electric vehicles to use the High Occupancy Vehicle (HOV) lanes and a differentiation in the tolls charged according to levels of contamination would help achieve these objectives. In 2009, London's mayor proposed reviewing the possibility that electric vehicles might use the HOV lanes, as introduced in the city of Oslo (Norway).

2) Establishment of "Park & Ride" sites at interchanges. Such car parking facilities, sited in city outskirts, would allow owners of electric vehicles to leave their cars charging while they are at work.

3) Agreements with private companies to install charging stations in firms' car parks. The report even considers the possibility that such charging points would serve as the point of association with private electric vehicles.

4) Financial support from the Institut Català d'Energia for the implementation of pilot programs that improve the energy efficiency of companies' transport systems. These companies receive up to 60% of the investment, provided they demonstrate an increase in energy efficiency of at least 5% in the displacement of workers from their homes to the workplace. Here, electric vehicles offer a competitive advantage in attaining this saving.

5) An information system designed for electric vehicle owners that allows them to quickly identify the nearest charging points. The introduction of clear, visible and uniform signs to help identify the recharge points is also useful in reducing the "anxiety" of the owners of electric vehicles.

For the US market, Skerlos and Winebrake (2010) also identify the use of HOV lanes as being one of the main measures for promoting sustainable mobility. In fact, as Diamond (2009) reports, implementing this measure increases the demand for hybrid vehicles in the United States. Consequently, it has already been introduced in the following states: Virginia, California, New York, Florida and Utah.

State	Date
California	August 10, 2005
Florida	October 1, 2005
New York	March 1, 2006
Utah	September 1, 2006
Virginia	June 30, 2006

Table 8: Permission for hybrid vehicles to use the HOV lanes in the US

Source: Diamond (2009)

#### 7. Other policies and their implications

In addition to the four main policy areas outlined above, there are other measures and factors that can influence the deployment of electric vehicles. The International Energy Agency (2008) has published a set of cross-section recommendations that should serve to promote the development, penetration and consolidation of electric vehicles. Market security is an essential element for vehicle and battery manufacturers to safeguard the investments required. It seems advisable, therefore, for government authorities to set transparent, medium-term goals and criteria so that private players in the industry can make efficient decisions.

The report also points out the need not to opt, as far as possible, for a given technology, but rather to fix set goals (e.g. a given reduction in CO2 emissions) and that it should be the private actors that determine which technology is the most efficient for achieving these objectives. The Government, together with industry, should even organize educational programs to increase consumer confidence and understanding of electric vehicles. Likewise, disseminating information about the operation of electric vehicles (including battery life, recharge times, location of charging points, type and cost of repairs, etc.) can help increase consumer trust in these vehicles.

A key element to consider in the development and subsequent market penetration of electric vehicles is product standardization. As Brown et al (2010) stress, standardization in the battery recharging infrastructure, electricity distribution and accounting for the environmental characteristics of this energy, and different features of smart grid technologies would help significantly in the introduction of electric vehicles. Good battery performance is also a key factor affecting the implementation of electric vehicles. In this regard, the U.S. Department of Energy's "FreedomCAR and Vehicle Technologies Program" promoted through the "Advanced Vehicle Testing Activity (AVTA)", tests the reliability and durability of batteries installed in hybrid electric vehicles or plug-in hybrids.

Karner and Francfort (2007) report outcomes from their analyses of the sets of batteries installed in such vehicles today. They show that battery life is significantly reduced in the presence of air conditioning and that some batteries fail before their theoretical life is reached (e.g. the Honda Insight battery begins to fail at 72,000 miles). They conclude that if consumers are to purchase these vehicles then they need to be

reassured that the batteries will perform and that ultimately they will be economically profitable.

Besides the benefits of electric vehicles and the economic policy measures that can help promote them, we need to take into account possible drawbacks resulting from their introduction. Indeed, the identification of the problems their deployment might generate should help minimize these and, eventually, ensure their success.

WWF (2008) reports that massive deployment of vehicle batteries will greatly increase the amount of raw materials required for their production. Moreover, the batteries will have to be properly recycled once they reach the end of their useful life cycle if serious environmental problems are to be averted.

A further element to consider is the impact that the introduction of electric vehicles might have on electricity production. As Hadley and Tsvetkova (2009) point out, if the recharging of hybrid vehicles is not undertaken solely during valley periods, then the increase in peak time demands could lead to a price hike. Indeed, any increase required in production capacity would significantly raise the costs of introducing the hybrid vehicle.<sup>13</sup>

This problem has also been noted in a study conducted in Catalonia (Spain) by the Generalitat de Catalunya (2009). The report proposes the creation of a fee "super valley" to serve as an incentive to the owners of electric vehicles so that they recharge their batteries at times of low electricity demand. Increasing consumption would not only require an increase in production capacity, but would necessitate a more efficient use of the currently installed capacity. Differentiating between day- and night-time electricity prices is also recommended by the report of the International Energy Agency (2008) as being essential in controlling the impact that electric vehicles will have on electricity consumption. The Transport & Environment (2009b) report reaches similar conclusions.

A final point to bear in mind is that the greater efficiency provided by electric vehicles should result in the increased share of road transport, which in turn could lead to increased net energy consumption.

<sup>&</sup>lt;sup>13</sup> See Green II et al (2011) for a detailed analysis of the impact of the introduction of plug-in hybrid vehicles on the electrical distribution network.

#### 8. Concluding remarks

Increasing fuel prices and growing environmental concerns are two key factors enhancing the potential of the electric vehicle as a valid alternative to the internal combustion engine. However, electric vehicles must still overcome a host of barriers (both technical and economic) if they are to compete with traditional vehicles.

The aim of this study has been to describe the main drivers of the development of electric vehicles, as well as the public and private policies that have been adopted, or might be applied, for promoting these vehicles as alternatives to ICE vehicles. In so doing we have analyzed the key points of reference in the field, undertaking a review of policies being implemented around the world.

We conclude that the development of an extensive charging network capable of overcoming the problems of "range anxiety"; the ability to guarantee sufficient demand for electric vehicles so as to maintain such charging networks; and the development of batteries (the principal component of an electric vehicle) that can provide greater autonomy while ensuring lower production and replacement costs are the goals that must be achieved if electric vehicles are to be successfully incorporated in the automobile market.

In overcoming these barriers, the role that the public sector plays will be crucial. Thus, most industrialized countries have implemented, in one form or another, public measures for just this purpose. These measures can be classed into four categories: first, the development of both public and private charging infrastructure or systems; second, traditional subsidies or tax breaks for the purchase of environmentally friendly vehicles (electric, hybrid, etc.); third, industrialization and R&D policies which, among other factors, reduce battery production costs (main expense in constructing these vehicles); and finally, the inclusion of electric vehicles in sustainable mobility programs. These are just some of the measures that public agencies can implement to help overcome the technological and economic barriers that limit the introduction and consolidation of electric vehicles.

The geographical spread of these measures is concentrated principally in the industrialized Northern Hemisphere, where Japan has most experience, not only in terms of the involvement of its public administration in implementing different strategies

of public funding, but also as regards its collaboration with the private sector to develop new technology (hybrid) batteries, etc. Elsewhere, the regulations introduced by the European Union and the measures taken by its member states, the United States and Canada represent the chief examples of attempts to develop the electric vehicle.

Today, many countries are implementing these, or similar, measures to facilitate the introduction and consolidation of the electric vehicle so that it might become the mode of transport of the future. However, the barriers remain considerable and greater involvement is required from the public administration to tackle "the chicken or the egg" dilemma faced by the sector and the negative effects that a poor tariff regulation would have on the electricity market.

#### References

- Ahman, M. (2001) "Primary energy efficiency of alternative powertrains in vehicles" *Energy*, Vol. 26, pp. 973-989.
- Ahman, M. (2006) "Government policy and development of electric vehicles in Japan" *Energy Policy*, 34(4), 433-443.
- Alberini, A., Harrington, W., and McConnell, V. (1995) "Determinants of participation in accelerated vehicle-retirement programs" *RAND Journal of Economics*, Vol. 26(1), pp. 93-112.
- Andersen, P.H., Mathews, J.A. and Ras, M. (2009) "Integrating private transport into renewable energy policy: the strategy of creating intelligent recharging grids for electric vehicles" *Energy Policy*, 37(7), 2481-2486.
- Beresteanu, A., and Li, S. (2011) "Gasoline prices, government support, and the demand for hybrid vehicles in the United States" *International Economic Review*, Vol. 52(1), pp. 161-182.
- Boston Consulting Group (2009) The comeback of the electric car? How real, how soon, and what must happen next. Boston, 2009.
- Bradley, Th.H., and Quinn, C.W. (2010) "Analysis of plug-in hybrid vehicle utility factor" *Journal* of *Power Sources*, Vol. 195, pp. 5399-5408.
- Brown, S., Pyke, D., and Steenhof, P. (2010) "Electric vehicles: The role and importance of standards in an emerging market" *Energy Policy*, Vol. 38, pp. 3797-3806.
- Busse, M., Silva-Risso, J., and Zettermeyer, F. (2006) "\$ 1000 cash back: The pass-through of auto manufacturer promotions" *American Economic Review*, Vol. 96(4), pp. 1253-1270.
- Chandra, A., Gulati, S., and Kandlikar, M. (2010) "Green drivers or free riders? An analysis of tax rebates for hybrid vehicles" *Journal of Environmental Economics and Management*, Vol. 60, pp. 78-93.
- Citi Investment Research & Analysis (2009) "All hail the electric car... but where will we plug them in?", Report. 23th September.
- Cooper, A., Chen, Y., and McAlinden, A. (2010) The economic and fiscal contributions of the "cash for clunkers" program – National and state effects. Center for Automotive Research (CAR) Research Memorandum. Washington.
- Diamond, D. (2009) "The impact of government incentives for hybrid-electric vehicles: Evidence from U.S. states" *Energy Policy*, Vol. 37, pp. 972-983.
- EVAAP, (2002). Newsletter Nº 20. January. Electrice Vehicle Association of Asia Pacific.
- Gallagher, K.S., and Muehlegger, E. (2011) "Giving green to get green? Incentives and consumer adoption of hybrid vehicle technology" *Journal of Environmental Economics and Management*, Vol. 61, pp. 1-15.
- Generalitat de Catalunya (2010) Estratègia d'impuls del vehicle elèctric a Catalunya. Barcelona.
- Green II, R.C., Wang, L., and Alam, M. (2011) "The impact of plug-in hybrid electric vehicles on distribution networks: A review and outlook" *Renewable and Sustainable Energy Reviews*, Vol. 15, pp. 544-553.
- Greenpeace, Friends of the Earth Europe and Transport & Environment (2010), Green power for electric cars. Harvesting the climate potential of electric vehicles. Brussels.

- Hadley, S.W., and Tsvetkova, A.A. (2009) "Potential impacts of plug-in hybrid electric vehicles on regional power generation" *The Electricity Journal*, Vol. 22(10), pp. 56-68.
- Hayashi, E., Takahashi, I., Funimasa, A. (1994) "Study impacts of electric vehicle charging to power supply systems", 12<sup>th</sup> electric vehicle symposium. December. Anaheim, CA, USA.
- Huang, E. (2010) "Do public subsidies sell green cars? Evidence from the U.S. "cash for clunkers" program" *Belfer Discussion Paper* No 17.
- Iguchi, M. (1992) "Market expansion programme of electric vehicles planned by the ministry of international trade and industry", Japan. In: Proceedings of the Urban Electric Vechile OECD-conference. Stockholm, 25-27 May.
- International Energy Agency (2008), Technology roadmap. Electric and plug-in hybrid electric vehicles. Paris.
- International Energy Agency-RETD (2010) "Opportunities for the use of renewable energy in road transport", Policy Makers Report, RETRANS, March.
- Jiménez, J.L., Perdiguero, J., and García, C. (2012a) "Evaluation of subsidies programs to sell green cars: Impact on prices, quantities and efficiency" *XREAP Working Paper*, No 2011-14.
- Jiménez, J.L., Perdiguero, J., and Voltes-Dorta, A. (2012b) "Are car manufacturers on the way to reduce CO2 emissions? A DEA approach" *Mimeo.*
- Kageson, P. (2005) Reducing CO2 emission from new cars: A progress report on the car industry's voluntary agreement and an assessment of the need for policy instrument. European Federation for Transport and Environment, Brussels.
- Karner, D., and Francfort, J. (2007) "Hybrid and plug-in hybrid vehicle performance testing by the UD Department Energy Advanced Vehicle Testing Activity" *Journal of Power Sources*, Vol. 174, pp. 69-75.
- Kley, F., Lerch, Ch., and Dallinger, D. (2011) "New business models for electric cars A holistic approach" *Energy Policy*, Vol. 39, pp. 3392-3403.
- KPMG (2010) "Sharing knowledge on topical issues in the automotive industry", KPMG International, Vol. Seven, October.
- Mayor of London (2009) "An electric vehicle delivery plan for London". May.
- Meier-Eisenmann, E., Moreni, G., and Simon, G. (2001) The utilization of recharging stations and reserved parking places. Swiss Energy.
- MITI (1990) Essentials of the electric vehicle market expansion programme in Japan. Ministry of International Trade and Industry Electric Vehicle Council. Tokyo.
- Offer, G.J., Howey, D., Contestabile, M., Clague, R., and Brandon, N.P. (2010) "Comparative analysis of battery electric, hydrogen fuel cell and hybrid vehicles in a future sustainable road transport system" *Energy Policy*, Vol. 38, pp. 24-29.
- Pike Research (2009), EV Consumer Survey. September.
- Romero, A. (2012) "Electric vehicle vs. conventional vehicle: economic valuation of environmental impacts". Presented at *VII Encuentro de la Asociación Española de la Economía Energética*. Navarra.
- Ryan, L., Ferreira, S., and Convery, F. (2009) "The impact of fiscal and other measures on new passenger car sales and CO<sub>2</sub> emissions intensity: Evidence from Europe" *Energy Economics*, Vol. 31, pp. 365-374.

- Transport and Environment (2009a) How to avoid and electric shock. Electric cars: From the hype to reality. Brussels.
- Transport and Environment (2009b), A sustainable future for transport. Response to the European Commission Consultation. Brussels.

Transport and Environment (2009c), Cut out EU loopholes to avoid an electric shock. Brussels.

Transport and Environment (2010), Electric cars will only work with 'green' power. Brussels.

- Sivak, M., and Schoettle, B. (2009) "The effect of "cash for clunkers" program on the overall fuel economy of purchased new vehicles" *Working Paper University of Michigan Transportation Research Institute* No 34.
- Skerlos, S.J., and Winebrake, J.J. (2010) "Targeting plug-in hybrid electric vehicle policies to increase social benefits" *Energy Policy*, Vol. 38, pp. 705-708.
- Srivastava, A.K., Annabathina, B., and Kamalasadan, S. (2010) "The challenges and policy options for integrating plug-in hybrid electric vehicle into the electric grid" *Electricity Journal*, Vol. 23(3), pp. 83-91.
- Sterner, Th. (2007) "Fuel taxes: An important instrument for climate policy" *Energy Policy*, Vol. 35, pp. 3194-3202.
- United States Department of Energy (2011), Strategic Plan. Washington.
- Weinert, J., Ogden, J., Sperling, D. and Burke, A. (2008) "The future of electric two-wheelers and electric vehicles in China" *Energy Policy*, 36(4), 2544-2555.
- Wiederer, A., and Philip, R. (2010) Policy options for electric vehicle charging infrastructure in C40 cities. Transportation, Clinton Climate Initiative. Boston.

World Wide Fund for Nature (WWF) (2008), Plugged in: The end of oil age. Brussels.

Yacobucci, B.D., and Canis, B. (2010) Accelerated vehicle retirement for fuel economy: "cash for Clunkers", Congressional Research Service Report for Congress. Washington.

#### Llista Document de Treball

List Working Paper

- WP 2012/08 "Policy options for the promotion of electric vehicles: a review" Perdiguero, J. and Jiménez, J.L.
- WP 2012/07 "Price differences between domestic and international air markets: an empirical application to routes from Gran Canaria" Fageda, X.; Jiménez, J.L. and Díaz Santamaría, C.
- WP 2012/06 "Building a "quality in work" index in Spain" López-Tamayo, J.; Royuela, V. and Suriñach, J.
- WP 2012/05 "Mergers and difference-in-difference estimator: why firms do not increase prices?" Jiménez, J.L. and Perdiguero, J.
- WP 2012/04 "What attracts knowledge workers? The role of space, social connections, institutions, jobs and amenities" Miguélez, E. and Moreno, R.
- WP 2012/03 "What Drives the Urban Wage Premium? Evidence along the Wage Distribution" Matano, A. and Naticchioni, P.
- WP 2012/02 "Location Patterns of Creative Capital and Regional Disparities in Spain" Kerimoglu, E. and Karahasan, B.C.
- WP 2012/01 "The connection between distortion risk measures and ordered weighted averaging operators" Belles-Sampera, J.; Merigó, J.M.; Guillén, M. and Santolino, M.
- WP 2011/26 "Productivity and innovation spillovers: Micro evidence from Spain" Goya, E.; Vayá, E. and Suriñach, J.
- WP 2011/25 "The regional distribution of unemployment. What do micro-data tell us?" López-Bazo, E. and Motellón, E.
- WP 2011/24 "Vertical relations and local competition: an empirical approach" Perdiguero, J.
- WP 2011/23 "Air services on thin routes: Regional versus low-cost airlines" Fageda, X. and Flores-Fillol, R.
- WP 2011/22 "Measuring early childhood health: a composite index comparing Colombian departments" Osorio, A.M.; Bolancé, C. and Alcañiz, M.
- WP 2011/21 "A relational approach to the geography of innovation: a typology of regions" Moreno, R. and Miguélez, E.
- WP 2011/20 "Does Rigidity of Prices Hide Collusion?" Jiménez, J.L and Perdiguero, J.
- WP 2011/19 "Factors affecting hospital admission and recovery stay duration of in-patient motor victims in Spain" Santolino, M.; Bolancé, C. and Alcañiz, M.
- WP 2011/18 "Why do municipalities cooperate to provide local public services? An empirical analysis" Bel, G.; Fageda, X. and Mur, M.
- WP 2011/17 "The "farthest" need the best. Human capital composition and development-specific economic growth" Manca, F.
- WP 2011/16 "Causality and contagion in peripheral EMU public debt markets: a dynamic approach" Gómez-Puig, M. and Sosvilla-Rivero, S.
- WP 2011/15 "The influence of decision-maker effort and case complexity on appealed rulings subject to multi-categorical selection" Santolino, M. and Söderberg, M.
- WP 2011/14 "Agglomeration, Inequality and Economic Growth: Cross-section and panel data analysis" Castells, D.

WP 2011/13 "A correlation sensitivity analysis of non-life underwriting risk in solvency capital requirement estimation" Bermúdez, L.; Ferri, A. and Guillén, M. WP 2011/12 "Assessing agglomeration economies in a spatial framework with endogenous regressors" Artis, M.J.; Miguélez, E. and Moreno, R. WP 2011/11 "Privatization, cooperation and costs of solid waste services in small towns" Bel, G; Fageda, X. and Mur, M. WP 2011/10 "Privatization and PPPS in transportation infrastructure: Network effects of increasing user fees" Albalate, D. and Bel, G. WP 2011/09 "Debating as a classroom tool for adapting learning outcomes to the European higher education area" Jiménez, J.L.; Perdiguero, J. and Suárez, A. WP 2011/08 "Influence of the claimant's behavioural features on motor compensation outcomes" Ayuso, M; Bermúdez L. and Santolino, M. WP 2011/07 "Geography of talent and regional differences in Spain" Karahasan, B.C. and Kerimoalu E. WP 2011/06 "How Important to a City Are Tourists and Daytrippers? The Economic Impact of Tourism on The City of Barcelona" Murillo, J; Vayá, E; Romaní, J. and Suriñach, J. WP 2011/05 "Singling out individual inventors from patent data" Miguélez, E. and Gómez-Miguélez, I. WP 2011/04 ¿La sobreeducación de los padres afecta al rendimiento académico de sus hijos?" Nieto, S; Ramos, R. WP 2011/03 "The Transatlantic Productivity Gap: Is R&D the Main Culprit?" Ortega-Argilés, R.; Piva, M.; and Vivarelli, M. "The Spatial Distribution of Human Capital: Can It Really Be Explained by WP 2011/02 Regional Differences in Market Access?" Karahasan, B.C. and López-Bazo, E WP 2011/01 "If you want me to stay, pay". Claeys, P and Martire, F WP 2010/16 "Infrastructure and nation building: The regulation and financing of network transportation infrastructures in Spain (1720-2010)"Bel,G "Fiscal policy and economic stability: does PIGS stand for Procyclicality In WP 2010/15 Government Spending?" Maravalle, A ; Claeys, P. WP 2010/14 "Economic and social convergence in Colombia" Royuela, V; Adolfo García, G. WP 2010/13 "Symmetric or asymmetric gasoline prices? A meta-analysis approach" Perdiguero, J. WP 2010/12 "Ownership, Incentives and Hospitals" Fageda, X and Fiz, E. "Prediction of the economic cost of individual long-term care in the Spanish WP 2010/11 population" Bolancé, C; Alemany, R; and Guillén M WP 2010/10 "On the Dynamics of Exports and FDI: The Spanish Internationalization Process" Martínez-Martín, J. WP 2010/09 "Urban transport governance reform in Barcelona" Albalate, D; Bel, G and Calzada, J. WP 2010/08 "Cómo (no) adaptar una asignatura al EEES: Lecciones desde la experiencia comparada en España" Florido C. ; Jiménez JL. and Perdiguero J. WP 2010/07 "Price rivalry in airline markets: A study of a successful strategy of a network carrier against a low-cost carrier" Fageda, X ; Jiménez J.L. ; Perdiguero , J. WP 2010/06 "La reforma de la contratación en el mercado de trabajo: entre la flexibilidad y la seguridad" Royuela V. and Manuel Sanchis M.

- WP 2010/05 "Discrete distributions when modeling the disability severity score of motor victims" Boucher, J and Santolino, M
- WP 2010/04 "Does privatization spur regulation? Evidence from the regulatory reform of European airports . Bel, G. and Fageda, X."
- WP 2010/03 "High-Speed Rail: Lessons for Policy Makers from Experiences Abroad". Albalate, D ; and Bel, G."
- WP 2010/02 "Speed limit laws in America: Economics, politics and geography". Albalate, D ; and Bel, G."
- WP 2010/01 "Research Networks and Inventors' Mobility as Drivers of Innovation: Evidence from Europe" Miguélez, E. ; Moreno, R. "
- WP 2009/26 "Social Preferences and Transport Policy: The case of US speed limits" Albalate, D.
- WP 2009/25 "Human Capital Spillovers Productivity and Regional Convergence in Spain", Ramos, R ; Artis, M.; Suriñach, J.
- WP 2009/24 "Human Capital and Regional Wage Gaps", López-Bazo, E. Motellón E.
- WP 2009/23 "Is Private Production of Public Services Cheaper than Public Production? A meta-regression analysis of solid waste and water services" Bel, G.; Fageda, X.; Warner. M.E.
- WP 2009/22 "Institutional Determinants of Military Spending" Bel, G., Elias-Moreno, F.
- WP 2009/21 "Fiscal Regime Shifts in Portugal" Afonso, A., Claeys, P., Sousa, R.M.
- WP 2009/20 "Health care utilization among immigrants and native-born populations in 11 European countries. Results from the Survey of Health, Ageing and Retirement in Europe" Solé-Auró, A., Guillén, M., Crimmins, E.M.
- WP 2009/19 "La efectividad de las políticas activas de mercado de trabajo para luchar contra el paro. La experiencia de Cataluña" Ramos, R., Suriñach, J., Artís, M.
- WP 2009/18 "Is the Wage Curve Formal or Informal? Evidence for Colombia" Ramos, R., Duque, J.C., Suriñach, J.
- WP 2009/17 "General Equilibrium Long-Run Determinants for Spanish FDI: A Spatial Panel Data Approach" Martínez-Martín, J.
- WP 2009/16 "Scientists on the move: tracing scientists' mobility and its spatial distribution" Miguélez, E.; Moreno, R.; Suriñach, J.
- WP 2009/15 "The First Privatization Policy in a Democracy: Selling State-Owned Enterprises in 1948-1950 Puerto Rico" Bel, G.
- WP 2009/14 "Appropriate IPRs, Human Capital Composition and Economic Growth" Manca, F.
- WP 2009/13 "Human Capital Composition and Economic Growth at a Regional Level" Manca, F.
- WP 2009/12 "Technology Catching-up and the Role of Institutions" Manca, F.
- WP 2009/11 "A missing spatial link in institutional quality" Claeys, P.; Manca, F.
- WP 2009/10 "Tourism and Exports as a means of Growth" Cortés-Jiménez, I.; Pulina, M.; Riera i Prunera, C.; Artís, M.
- WP 2009/09 "Evidence on the role of ownership structure on firms' innovative performance" Ortega-Argilés, R.; Moreno, R.

WP 2009/08	"¿Por qué se privatizan servicios en los municipios (pequeños)? Evidencia empírica sobre residuos sólidos y agua" Bel, G.; Fageda, X.; Mur, M.
WP 2009/07	"Empirical analysis of solid management waste costs: Some evidence from Galicia, Spain" Bel, G.; Fageda, X.
WP 2009/06	"Intercontinental fligths from European Airports: Towards hub concentration or not?" Bel, G.; Fageda, X.
WP 2009/05	"Factors explaining urban transport systems in large European cities: A cross- sectional approach" Albalate, D.; Bel, G.
WP 2009/04	"Regional economic growth and human capital: the role of overeducation" Ramos, R.; Suriñach, J.; Artís, M.
WP 2009/03	"Regional heterogeneity in wage distributions. Evidence from Spain" Motellón, E.; López-Bazo, E.; El-Attar, M.
WP 2009/02	"Modelling the disability severity score in motor insurance claims: an application to the Spanish case" Santolino, M.; Boucher, J.P.
WP 2009/01	"Quality in work and aggregate productivity" Royuela, V.; Suriñach, J.
WP 2008/16	"Intermunicipal cooperation and privatization of solid waste services among small municipalities in Spain" Bel, G.; Mur, M.
WP 2008/15	"Similar problems, different solutions: Comparing refuse collection in the Netherlands and Spain" Bel, G.; Dijkgraaf, E.; Fageda, X.; Gradus, R.
WP 2008/14	"Determinants of the decision to appeal against motor bodily injury settlements awarded by Spanish trial courts" Santolino, ${\rm M}$
WP 2008/13	"Does social capital reinforce technological inputs in the creation of knowledge? Evidence from the Spanish regions" Miguélez, E.; Moreno, R.; Artís, M.
WP 2008/12	"Testing the FTPL across government tiers" Claeys, P.; Ramos, R.; Suriñach, J.
WP 2008/11	"Internet Banking in Europe: a comparative analysis" Arnaboldi, F.; Claeys, P.
WP 2008/10	"Fiscal policy and interest rates: the role of financial and economic integration" Claeys, P.; Moreno, R.; Suriñach, J.
WP 2008/09	"Health of Immigrants in European countries" Solé-Auró, A.; M.Crimmins, E.
WP 2008/08	"The Role of Firm Size in Training Provision Decisions: evidence from Spain" Castany, L.
WP 2008/07	"Forecasting the maximum compensation offer in the automobile BI claims negotiation process" Ayuso, M.; Santolino, M.
WP 2008/06	"Prediction of individual automobile RBNS claim reserves in the context of Solvency II" Ayuso, M.; Santolino, M.
WP 2008/05	"Panel Data Stochastic Convergence Analysis of the Mexican Regions" Carrion- i-Silvestre, J.L.; German-Soto, V.
WP 2008/04	"Local privatization, intermunicipal cooperation, transaction costs and political interests: Evidence from Spain" Bel, G.; Fageda, X.
WP 2008/03	"Choosing hybrid organizations for local services delivery: An empirical analysis of partial privatization" Bel, G.; Fageda, X.
WP 2008/02	"Motorways, tolls and road safety. Evidence from European Panel Data" Albalate, D.; Bel, G.
WP 2008/01	"Shaping urban traffic patterns through congestion charging: What factors drive success or failure?" Albalate, D.; Bel, G.
WP 2007/19	"La distribución regional de la temporalidad en España. Análisis de sus determinantes" Motellón, E.

- WP 2007/18 "Regional returns to physical capital: are they conditioned by educational attainment?" López-Bazo, E.; Moreno, R.
- WP 2007/17 "Does human capital stimulate investment in physical capital? evidence from a cost system framework" López-Bazo, E.; Moreno, R.
- WP 2007/16 "Do innovation and human capital explain the productivity gap between small and large firms?" Castany, L.; López-Bazo, E.; Moreno, R.
- WP 2007/15 "Estimating the effects of fiscal policy under the budget constraint" Claeys, P.
- WP 2007/14 "Fiscal sustainability across government tiers: an assessment of soft budget constraints" Claeys, P.; Ramos, R.; Suriñach, J.
- WP 2007/13 "The institutional vs. the academic definition of the quality of work life. What is the focus of the European Commission?" Royuela, V.; López-Tamayo, J.; Suriñach, J.
- WP 2007/12 "Cambios en la distribución salarial en españa, 1995-2002. Efectos a través del tipo de contrato" Motellón, E.; López-Bazo, E.; El-Attar, M.
- WP 2007/11 "EU-15 sovereign governments' cost of borrowing after seven years of monetary union" Gómez-Puig, M..
- WP 2007/10 "Another Look at the Null of Stationary Real Exchange Rates: Panel Data with Structural Breaks and Cross-section Dependence" Syed A. Basher; Carrion-i-Silvestre, J.L.
- WP 2007/09 "Multicointegration, polynomial cointegration and I(2) cointegration with structural breaks. An application to the sustainability of the US external deficit" Berenguer-Rico, V.; Carrion-i-Silvestre, J.L.
- WP 2007/08 "Has concentration evolved similarly in manufacturing and services? A sensitivity analysis" Ruiz-Valenzuela, J.; Moreno-Serrano, R.; Vaya-Valcarce, E.
- WP 2007/07 "Defining housing market areas using commuting and migration algorithms. Catalonia (Spain) as an applied case study" Royuela, C.; Vargas, M.
- WP 2007/06 "Regulating Concessions of Toll Motorways, An Empirical Study on Fixed vs. Variable Term Contracts" Albalate, D.; Bel, G.
- WP 2007/05 "Decomposing differences in total factor productivity across firm size" Castany, L.; Lopez-Bazo, E.; Moreno, R.
- WP 2007/04 "Privatization and Regulation of Toll Motorways in Europe" Albalate, D.; Bel, G.; Fageda, X.
- WP 2007/03 "Is the influence of quality of life on urban growth non-stationary in space? A case study of Barcelona" Royuela, V.; Moreno, R.; Vayá, E.
- WP 2007/02 "Sustainability of EU fiscal policies. A panel test" Claeys, P.
- WP 2007/01 "Research networks and scientific production in Economics: The recent spanish experience" Duque, J.C.; Ramos, R.; Royuela, V.
- WP 2006/10 "Term structure of interest rate. European financial integration" Fontanals-Albiol, H.; Ruiz-Dotras, E.; Bolancé-Losilla, C.
- WP 2006/09 "Patrones de publicación internacional (ssci) de los autores afiliados a universidades españolas, en el ámbito económico-empresarial (1994-2004)" Suriñach, J.; Duque, J.C.; Royuela, V.
- WP 2006/08 "Supervised regionalization methods: A survey" Duque, J.C.; Ramos, R.; Suriñach, J.
- WP 2006/07 "Against the mainstream: nazi privatization in 1930s germany" Bel, G.
- WP 2006/06 "Economía Urbana y Calidad de Vida. Una revisión del estado del conocimiento en España" Royuela, V.; Lambiri, D.; Biagi, B.

Institut de Recerca en Economia Aplicada Regional i Pública Research Institute of Applied Economics

- WP 2006/05 "Calculation of the variance in surveys of the economic climate" Alcañiz, M.; Costa, A.; Guillén, M.; Luna, C.; Rovira, C.
- WP 2006/04 "Time-varying effects when analysing customer lifetime duration: application to the insurance market" Guillen, M.; Nielsen, J.P.; Scheike, T.; Perez-Marin, A.M.
- WP 2006/03 "Lowering blood alcohol content levels to save lives the european experience" Albalate, D.
- WP 2006/02 "An analysis of the determinants in economics and business publications by spanish universities between 1994 and 2004" Ramos, R.; Royuela, V.; Suriñach, J.
- WP 2006/01 "Job losses, outsourcing and relocation: empirical evidence using microdata" Artís, M.; Ramos, R.; Suriñach, J.



## 

Institut de Recerca en Economia Aplicada Regional i Pública Research Institute of Applied Economics

**Universitat de Barcelona** Av. Diagonal, 690 • 08034 Barcelona

WEBSITE: www.ub.edu/irea/ • CONTACT: irea@ub.edu

